Contents lists available at ScienceDirect



Renewable and Sustainable Energy Reviews



journal homepage: www.elsevier.com/locate/rser

# Factors influencing the adoption of zero-emission buses: A review-based framework

Alessandro Avenali, Giuseppe Catalano, Mirko Giagnorio<sup>\*</sup>, Giorgio Matteucci

Department of Computer, Control and Management Engineering, Sapienza University of Rome, Italy

# ARTICLE INFO

Keywords: Barriers Electric buses Decision-making Enablers Public transportation Zero-emission buses

# ABSTRACT

Zero-emission buses (ZEBs) currently face the challenge of switching from small-scale implementation to largescale deployment. This stage is affected by multiple decision-making factors that either hinder or enable the ZEB adoption, also depending on local contexts and stakeholder perspectives. This study develops a holistic framework relating the transition towards ZEB fleets based on an extensive literature review. The review aims at fully capturing the supporting and limiting factors for the bus fleet decarbonisation and organising them in a comprehensive taxonomy and hierarchy. Subsequently, the impact of each of the identified factors was quantitatively analysed. The results show that technological, economic and managerial factors have hampered the large-scale deployment of ZEBs, while social, environmental, and institutional dimensions have stimulated their diffusion. However, more recent studies show that advances in the technological and managerial domains are reducing some drawbacks linked to ZEBs. Finally, the proposed framework is used to identify factors of growing significance that merit further research.

# 1. Introduction

The transport sector is currently one of the most significant contributors to climate change, playing a pivotal role in aligning urban economics with the global Sustainable Development Goals (SDGs). To address this issue, the transition towards cleaner power supply modes is gathering speed, considering both private vehicles and commercial fleets [1,2]. At present, electric powertrains are the only technology able to nullify harmful emissions from the tailpipe, using the energy provided by an electric battery or a fuel cell (i.e., hydrogen-feed). Although the environmental impact of these technologies (from a life-cycle perspective) depends on electricity and hydrogen generation mix and on the production and disposal of their components [3,4], they are also known as zero-emission vehicles.

In this context, public-funded bus services are ideal for testing the effects of transitioning from the early adoption to the early majority diffusion of these new technologies [5], and crossing the "chasm" to market penetration described by Berkeley et al. [6] for electric vehicles. Indeed, certain characteristics of public transport services make them particularly well-suited to electric powertrains. First, transport providers operate with fixed routes and schedules, centralised depots and shared infrastructures [7], thereby limiting the impact of any drawbacks

linked to electric battery range. The high yearly mileage of transit buses brings more leverage to the lower running costs of an electric powertrain compared to fossil fuelled options [8]. In addition, regulated utilities are more subject to public scrutiny [9], and thereby often subject to obligations and direct incentives introduced by national and local governments. In this regard, the high visibility of urban buses may help to raise public awareness of the importance of fighting air pollution, climate change and fossil fuel imports [10]. Finally, zero-emission buses (ZEBs), in addition to directly reducing harmful emissions, may indirectly improve air quality, habitat damage, congestion and noise within cities by influencing a modal shift from single-occupant vehicles to public transport. Indeed, some travel surveys have found a higher willingness to use a bus if it is electric [11,12], even at a higher fare (for people deeply concerned about environmental issues, such as young people) [13]. For these reasons, bus fleet decarbonisation strategically aligns with different SDGs. Notably, the explicit inclusion of public transport expansion in SDG target 11.2 (Affordable and sustainable transport systems) underscores its significance, while ZEBs also directly contribute to SDG 13 (Climate Action) by mitigating greenhouse gas emissions and air pollution.

Accordingly, governments worldwide are boosting public policies to promote ZEB deployment [1]. To such an extent that certain predictions, particularly in urban areas, anticipate a complete shift to electric city

\* Corresponding author. *E-mail address:* mirko.giagnorio@uniroma1.it (M. Giagnorio).

https://doi.org/10.1016/j.rser.2024.114388

Received 8 April 2023; Received in revised form 19 January 2024; Accepted 24 March 2024 Available online 31 March 2024

<sup>1364-0321/© 2024</sup> The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

# List of abbreviations

including units and nomenclature Greenhouse Gas Emissions GHG Sustainable Development Goals SDGs Zero-emission buses ZEBs

buses within the current decade [14].

Despite these advantages, the market penetration of ZEBs is not significantly larger than that of private electric vehicles. In the EU in 2021, full electric cars accounted for 9.1% of all new registrations, similar to electric buses, which accounted for 10.1% [15,16]. Worryingly, while diesel cars represent less than 20% of new car registrations, conventional diesel buses still account for more than 60%.

An extensive literature has addressed the consumer preferences for electric vehicles, and the literature presents various reviews of the factors influencing electric vehicle adoption (e.g., Refs. [17–23]). While ZEBs have received less attention overall [24], the number of studies focusing on electric buses has exponentially grown over the last 5 years [7,25,26]. The most frequent topics addressed in the literature with respect to electric buses are operations management (e.g., energy consumption, charging scheduling), cost-benefit analyses, and environmental and social aspects. More recently, the issue of charging infrastructure planning for transport electrification has been analysed by considering both private vehicles and public fleets [27].

In general, several studies have dealt with the decision-making process related to ZEB adoption, which involves public transport authorities, transport operators and other stakeholders of growing importance (e.g., Refs. [28–31]).

Given the burgeoning literature exploring the transition to a cleaner public transport, this research aimed at fully capturing the debate over the barriers to and enablers of ZEB adoption. Building on an extensive and up-to-date review of the literature on alternatively powered buses, it is proposed a holistic framework for mapping and assessing the decisionmaking factors that influence the large-scale deployment of ZEBs. To this end, the entire transition process is considered without adopting a specific perspective related to one of the multiple actors involved. Indeed, the limiting or enabling effect of each factor may change, depending on stakeholder standpoints and local contexts. This underlines the importance of collaborative governance for bus fleet renewal projects, as already highlighted in the literature [32–36].

This study contributes to the literature in three respects. First, it defines a comprehensive framework that includes all of the decisionmaking factors identified in the literature as barriers to or enablers of the transition to a ZEB fleet, as well as a taxonomy for scientific research and practitioners approaching the study of bus fleet decarbonisation. Second, it provides a quantitative analysis of the influence direction (hindering or enabling) of each factor, based on a systematic comparison of the case studies presented in the literature. It also investigates the time trend effects of technological and managerial progress on these factors. Finally, it identifies factors of growing significance that merit further research.

The paper is organised as follows: Section 2 summarises the methodology used to collect and analyse the relevant studies; Section 3 describes the ZEB fleet transition framework; Section 4 presents the quantitative results and findings; and Section 5 provides concluding remarks.

# 2. Method

# 2.1. Search strategy and study selection

This research is based on an extensive and up-to-date review of the

literature on the barriers to and enablers of bus fleet decarbonisation, aimed at identifying and mapping decision-making factors affecting the large-scale development of ZEBs. A literature search was conducted on the Scopus database, using 31 keywords reported in Table 1.

The research queries were based on three main sets of keywords, combined through the Boolean operator AND, namely: the first group related to the bus technology taxonomy, the second focused on decisionmaking factors (e.g., barriers, motivators, challenges, innovation, etc.), and the third limited the analysis to the bus sector. The search was restricted to article titles and keywords, in order to maintain the strict scope of the study. To ensure that all relevant studies on the barriers to and enablers of the ZEB transition were included in the review, the sample of papers was further expanded by using the snowballing literature review method. This enhanced the validity of the review with reference to backward and forward citations [37]. The most relevant papers were those discussing the ZEB adoption process and those identifying challenges and opportunities for transport operators, public transport agencies, and other stakeholders. While this analysis focused on the bus sector, it also includes major studies dealing with the diffusion of alternative fuel vehicles within industrial fleets (e.g., general heavy-duty vehicles, commercial fleets). It is worth noting that barriers and enablers relating bus fleet decarbonisation were often investigated without any differentiation between power technologies; indeed, many studies referred broadly to alternative fuel powertrains (e.g., Refs. [28, 38]). This study focuses on so-called zero-emission buses (i.e., battery electric buses and fuel cell buses), which are able to nullify the pollution produced from the tailpipe (in the tank-to-wheel stage).

The final set of papers included 47 peer-reviewed journal articles, 2 papers published in conference proceedings and 1 book chapter. All papers were published in the period 1997–2022, as the publication search window ended in December 2022.

# 2.2. Content analysis and data collection

The research was conducted by carrying out a two-step content analysis of the selected articles, as shown in Fig. 1.

First, to identify the decision-making factors and facilitate comparison, each paper was classified according to a grid highlighting the main relevant aspects. In particular, information was gathered regarding the study objectives, methods, geographical location of the case studies, and power technologies under review. A content analysis was conducted using inductive reasoning, as this approach is considered particularly effective for analysing innovation diffusion processes [38].

While this context-sensitive exploration of the influencing factors offers a comprehensive understanding of identified challenges and opportunities in the literature, it suffers the potential for subjectivity in the interpretation of findings. The human-conducted collection of an extensive array of factors, based on the review of numerous case studies, proved to be time-consuming; however, it was essential for ensuring the

# Table 1

List of the keywords used in the automatic literature search.

Power technologies	Technology adoption		Bus sector
Zero-emission	Decision- mak*	Trend	Bus
Electric	Factor	Adoption	Transit
Electrification	Barrier	Acceptance	Public
			transport*
Fuel cell	Deterrent	Criteria	
Hydrogen	Motivator	Transition	
Biogas	Driver	Purchase	
Alternative fuel	Challenge	Procurement	
		Innovation	
Biofuel Renewable Natural gas	Enabler Opportunity Tendency		



Fig. 1. Content analysis workflow.

identification of decision-making factors grounded in a contextual and critical understanding. The absence of a specific perspective related to any single actor in the framework adds another layer of complexity, as it requires reconciling the varied interests and priorities of multiple stakeholders. Notably, the proposed taxonomy has to recognise that factors may function as both barriers and enablers depending on the stakeholder standpoint and local context. This approach, applied consistently across the entire framework, guarantees its flexibility in addressing different regional and situational nuances. For example, the impact of well-to-wheel emissions of electric buses was found to vary depending on the electricity production methods in a given country [39, 40]. From a policy-making perspective, in regions with a high share of renewable energy, this factor acts as an enabler, whereas in contexts relying on fossil energy sources, it serves as a deterrent to ZEBs promotion. Then the framework can be effectively applied to different local contexts by calibrating the influence direction (i.e., hindering or enabling) and the scale of impact of each factor according to peculiar characteristics of the case study.

Second, the identified decision-making factors of the proposed framework were quantitatively analysed. This analysis was based on a deeper second reading of the articles to systematically gather data about deterring and facilitating factors identified in each study. Through this process, it was obtained a dataset of 768 barriers to and enablers of ZEB adoption, with each attached to one or more components of the proposed framework. The built database formed the basis for calculating descriptive statistics on the frequency with which each factor was mentioned in the different case studies, also considering differences linked to the time of publication. In this way, this research provides a detailed overview of how challenges and opportunities associated with the large-scale deployment of ZEBs have been explored in the existing literature.

# 3. ZEB fleet transition framework

This section introduces the holistic framework that was developed to define the comprehensive taxonomy of the decision-making factors affecting ZEB adoption. The framework displays a hierarchical structure, consistent with other models developed for multi-criteria decision-making in transport and energy sectors [2,41–43].

The first hierarchical level consists of six evaluative dimensions: (i) technological, (ii) economic, (iii) managerial, (iv) social, (v) environmental and (vi) institutional. Four of these are inspired by the prism of sustainability [44], which extends the well-known *three pillars approach* [45] to the institutional sphere. The technological and managerial

dimensions relate to technology adoption theories, with respect to innovation and environmental management. In this respect, it can be mentioned some seminal contributions: the *technology, organisation and environment* (TOE) *model* [46], the *unified theory of acceptance and use of technology* (UTAUT) [47] and the *attributes of innovation* identified by Rogers et al. [48]. These models also include non-technical aspects that steer innovation diffusion, which have been extensively investigated with respect to the consumer adoption of electric vehicles (e.g., Refs. [20,49,50]), but much less with respect to the mass-transit sector. Only a few studies have also considered this perspective with respect to alternatively fuelled buses (i.e. [10,28,30]).

The six evaluative dimensions are articulated in 15 decision-making factors, which, in turn, comprise 46 sub-factors that can positively or negatively impact the transition to ZEBs. Fig. 2 displays the hierarchical structure of the ZEB fleet transition framework.

The proposed framework is also interrelated with various Sustainable Development Goals (SDGs), in line with other frameworks focusing on environmental preservation [51]. Firstly, the enhancement of public transport aligns closely with SDG 11 (Sustainable Cities and Communities) by curbing pollution and alleviating congestion, as well as pursuing equity goals related to the level of accessibility and affordability of the transport systems. The identified factors within the environmental dimension seek to measure the contribution of ZEB transition to SDG 13 (Climate Action) by mitigating greenhouse gas emissions and climate change. Notably, the decision-making factor relating the life-cycle assessment of vehicles and infrastructure also considers the importance to ensure responsible consumption and production (SDG 12). The technological and managerial dimensions further align with SDG 9 (Industry, Innovation, and Infrastructure), as various stakeholders are interested in evaluating how cleaner bus services foster innovation in transportation technologies and support the development of sustainable energy supply infrastructures. Finally, the SDG 8 (Decent Work and Economic Growth) is included in the proposed framework through the social dimension, which is gaining increasing relevance also in modern climate action planning [52].

# 3.1. Technological dimension

The first evaluative dimension focuses on the impact of the power technology on the ability to effectively and efficiently provide transit services.

# 3.1.1. Technical and operational performance

Technical and operational performance refers to vehicle



Fig. 2. Hierarchical structure of the ZEB fleet transition framework.

characteristics, supporting infrastructures and compatibility with public transport services.

The first critical issue is the *driving range* limitation linked to battery electric buses, as mentioned in nearly every paper. This directly impacts vehicle flexibility and service scheduling, leading to a degradation of service (at the same cost) and additional operational complexity [53, 54]. However, some studies highlighted that the particular features of public transport (e.g., fixed routes and schedules, centralised depots, shared infrastructures) make transit electrification easier, relative to electrification in private mobility [7]. Manzolli et al. [25] noted that most recent energy management strategies (e.g., dynamic programming) can be efficiently implemented, thereby also improving the useful life of batteries. In general, advanced battery management will improve the effectiveness and efficiency of electric powertrains [55-57]. Aldenius et al. [32] analysed some cases in which opportunity-charging buses were introduced as a solution to the limited range. Hydrogen fuel cell buses increase vehicle range, while also serving as range extenders in hybrid solutions with electric batteries [58].

ZEB adoption poses additional challenges related to *charging/refuelling time*, which also increases operational complexity. Interestingly, there is a trade-off between bus frequency and charging needs, especially for opportunity charging options [59]. Nevertheless, new bus routing and scheduling tools, based on optimisation models, are needed to integrate bus service operations and charging infrastructures [29,60].

Some *maintenance issues* connected with the ZEB transition have also been reported. In particular, battery degradation and the useful life of batteries after the warranty period are concerns for bus fleet owners [59]. ZEBs thus require new safety schemes, which may also impact maintenance operations [29,61,62].

In terms of *energy efficiency*, electric powertrains outperform internal combustion engines, due to their higher engine yield [63]. However, the energy consumption of battery electric buses is more influenced by contextual factors (e.g., speed variation, weather conditions), which may reduce this technological advantage [64–66]. Moreover, bulky tanks or batteries decrease loading ability [54]. However, Thorne et al. [67] highlighted that there is only a small reduction in seat capacity due to the total mass limit of the bus, and this does not affect the load capacity when standing plots are considered.

The final factor identified with respect to the technical sphere was *compatibility with operational practices*. ZEBs require a different

operational approach, and the locked-in daily activities of transport providers may slow down the transition [68]. Bae et al. [30] found that investments in a specific fuel option tend to increase the likelihood that the same decision-maker will reject other alternative technologies. In this respect, fuel cell buses may represent a valid alternative, since they do not require frequent recharging and are more similar to conventional fossil fuels [61].

#### 3.1.2. Supporting infrastructures and energy/fuel supply

Supporting infrastructures and energy/fuel supply relate to the accessibility of recharging/refuelling stations, also considering depot upgrades to manage new on-site operations related to ZEBs.

The reliability, availability and quality of energy/fuel play a pivotal role in large-scale ZEB deployment. In this sense, transport planners face a set of challenges: the low capacity/willingness of the energy supply chain to deliver power promptly and at the agreed volume [33], grid congestion and priority allocation during electricity shortages [25,69], a limited ability to mass-produce low- or zero-carbon hydrogen [70] and other exogenous risks (e.g., trade wars in the energy/fuel supply chain, dependence on raw materials for electric vehicles). However, there are many industrial ways to reliably produce both green electricity and hydrogen that can be integrated into bus charging/refuelling strategies, such as photovoltaic solar generation [71] and electrolysis for hydrogen [61]. Manzolli et al. [25] highlighted that electrified bus fleets constitute "moving energy storage systems" that can contribute to grid resilience, as buffers to smooth out peak loads. Moreover, the use of ultracapacitors and other smart measures can decrease the maximum power demand [58,59]. Regarding more strategic considerations, Bae et al. [30] noted that the diversification of fuel technologies can alleviate operational risk. Additionally, Vodovozov et al. [61] suggested that several production plants of "green" hydrogen be designed for the coming years. Regarding the resilience of the transit service network, Hensher [72] emphasised the potential unreliability of a public transport system powered only by electricity, as, "if power goes down everything goes down ... trains, trams, buses".

ZEB feasibility also depends on the *availability of refuelling/charging infrastructure* (e.g., bus chargers, electric grid connections, hydrogen refuelling stations) for bus fleets. This is largely considered a barrier to ZEB adoption due to the challenges involved in installing chargers in dense urban areas (e.g., cable routing on public roads/pavements, power provision), finding spaces for on-site refuelling and maintenance, and meeting parking requirements linked to longer stops [29,70]. In this respect, public and private actors (e.g., transport and energy operators) must cooperate to plan alternative charging strategies in line with city and transit service [67]. While the availability of public infrastructure undoubtedly enables ZEB implementation, bus fleet operators may also rely on central facilities [73,74].

Bus *depot upgrades* are needed to accommodate ZEB infrastructure and operations. Civil work (e.g., excavation, conduits, cabling, repaving) may be required to bring power to bus parking locations, and this may delay implementation [58], over and above any local government approvals [72]. Furthermore, ZEBs require new safety and prevention standards, as health and safety concerns are critical to depot operation. In fact, a higher annual frequency of accidents has been observed in electrified bus fleets due to thermal runaways in battery packs [75,76]. Moreover, hydrogen tends to easily flow out of the tanks [61]. In general, ZEBs require new safety and prevention standards.

Finally, *after-sale support and assistance* for both vehicles and infrastructure may impact ZEB diffusion. In recent years, unknown and distant suppliers, shortages of spare parts and unreliable after-sale services have hindered the large-scale deployment of electric buses [18, 77]. However, warranties and maintenance provided by manufacturers have recently been found to facilitate the adoption of alternatively powered buses [30]. In addition, some service providers have started to sell turnkey solutions (i.e., "vehicle-as-a-service"), bundling bus maintenance, charging infrastructure and energy provision into a single charge to the operator [32,33].

#### 3.1.3. Service scheduling and fleet operations

Service scheduling and fleet operations may require ZEB technologies to adapt to current transit provision requirements and operations, or vice versa.

In this regard, service features and the geographical context may pose various challenges to ZEB diffusion. Indeed, demanding driving regimes related to route topology and bus transit characteristics (e.g., a hilly terrain, slow traffic, frequent stopping, a high load factor, weekly supply variation, extreme weather conditions) may significantly affect efficiency and effectiveness (see, e.g., Refs. [25,31,67,78]). In this regard, public space and urban density play influence electric traction feasibility, especially for trolleybus systems [79]. Conversely, several public transport characteristics match well with ZEB specifications, including fixed daily routes, long idle times for recharge, and small daily variance in the required duty cycles [80]. Studies have underlined the connections between context, battery type and charging strategy for electrified bus fleets [57,74,76]. Local areas with infrastructure available to generate renewable electricity and rural areas that may do so at a lower cost may support the transition to ZEB systems [59,71]. In this context, new computational capabilities related to bus operation planning and scheduling may also help to accelerate ZEB diffusion (e.g., predictive energy consumption management, decision support models) [7,29,81].

These optimisation techniques are key to facilitating effective *mixed fleet management*. Indeed, transport operators are currently challenged to handle different power/fuel sources simultaneously at depots. The integration of practices related to conventional fossil fuels and electric buses may increase management complexity, also considering the expertise of diesel-trained staff [33]. However, reliance on various power technologies may alleviate operational risk by determining the most effective alternatives for specific needs and taking advantage of dynamic pricing models for different power/fuel sources [58,82].

# 3.2. Economic dimension

The economic dimension focuses on the cost performance of ZEBs compared to fossil fuelled buses, focusing on CAPEX and OPEX expenses. Given that ZEB adoption is determined at an organisational level (by both public and private decision-makers, as defined by Biresselioglu

et al. [18]), it is crucial to carry out a financial assessment of ZEB technologies.

#### 3.2.1. CAPEX and other initial costs

CAPEX and other initial costs refer to all of the capital expenses related to upfront investments and fleet revamping. Regarding start-up expenses, decision-makers must also consider pre-purchase and transaction costs (e.g., the search for and selection of suppliers).

*Vehicle costs* include bus purchasing costs, midlife costs (e.g., battery replacement), and disposal costs. These costs are considered a significant barrier to ZEB adoption, as they are higher than the comparative costs for buses powered by conventional fossil fuels (see, e.g., Refs. [29, 83,84]). Battery costs still account for a considerable share of the vehicle costs [85], and their useful life is approximately half that of internal combustion buses [59]. However, some studies remarked that energy storage hybridisation (i.e., batteries, super-capacitors) can potentially reduce electric bus costs [25,86]. Fuel cell buses are comparatively expensive due to the rare substances needed for their construction, as well as their small-scale production [61].

The other significant investment related to ZEB fleets concerns *infrastructure costs*, including installing and running electric chargers and refuelling stations and modifying depots [87]. Blynn and Attanucci [53] identify potential diseconomies of scale related to electric bus infrastructures, as chargers are often bundled with the vehicle (i.e., electricity upgrades, more depot space).

A thorough economic assessment of ZEBs must also consider *indirect procurement costs*, referring to all expenses and activities that support the effective deployment of the power technology. These are influenced by a lack of brand and model variety, a lack of competition (i.e., higher negotiation costs), the long time needed to develop fruitful collaboration with suppliers, and a lack of operational capabilities [30,32,63]. In many countries, competition among bus manufacturers and energy retailers has been growing, with a positive impact for ZEB fleet development [33].

#### 3.2.2. Operational costs

In this framework, OPEX refers to variable costs connected with bus service provision, mainly reflected in fuel/energy and maintenance expenses.

*Fuel/energy costs* differ substantially between battery electric buses and fuel cell buses. In the first case, they are widely considered an enabler to adoption, due to the lower price of electricity compared to other fuel alternatives [18]. Moreover, transport providers can lever the dynamic pricing of electricity (e.g., lower charges in off-peak periods) by adopting real-time charging scheduling systems, thereby significantly reducing operational costs [7]. The energy demand charge must not be underestimated, since it can significantly increase operating expenses and, especially if load cannot be shifted to off-peak periods, discourage bus fleet electrification [58]. In this context, the implementation of distributed energy generation hubs (mainly powered by solar energy) may cut costs related to grid usage [25]. Conversely, the cost of hydrogen is a key barrier to fuel cell bus diffusion, due to both production and retail expenses [70].

*Maintenance costs* are quite variable and depend on various subfactors, including the choice of whether to outsource services. Some studies have highlighted maintenance cost savings related to electric bus adoption as a consequence of the reduced number of components (e.g., no emission control equipment or oil changes) [30]. Fuel cell buses, instead, suffer from limited development [83].

# 3.2.3. Financial evaluations

Financial evaluations refer to the impact of ZEBs on the financial performance of fleet owners, considering potential returns on investment and public transport revenues, as well as financial risks.

*Financial risks and returns on investment* are difficult to predict, since the overall costs incurred by switching to ZEBs are uncertain [32].

However, fleet owners must consider risks associated with over-investing in a particular technology under development (as it could be soon obsolete), sunk costs related to asset-intensive technologies (e. g., ZEBs), cost escalation in contracts, and public funding uncertainty [33,54].

In this context, in addition to the crucial role of public funding (see Section 3.6), the lack of start-up capital might be addressed through *debt instruments*. Particular bonds and soft loans (e.g., green bonds, concessional loans, international climate loans) may help to reduce the cost of financing [77]. Furthermore, long-term solutions may allow third-parties to provide capital assets through leasing schemes [18,67].

ZEB adoption may also impact *public transport revenues*, as some studies have found a positive consumer intention to use electric buses over and above fossil fuelled buses [7]. However, studies have produced mixed findings on whether consumers are willing to pay more to use ZEBs [29,80]. In general, low public transport service revenues are likely to delay investment [87], and it may be necessary to reduce the bus supply in order to fund new ZEB technologies [53]. Recent studies remarked the importance of improving public transport quality and studying users' perceptions [88].

# 3.3. Managerial dimension

The managerial dimension analyses how organisational characteristics might influence the transition to a ZEB fleet, also considering levels of knowledge about the relevant technologies.

#### 3.3.1. Organisational structure and culture

Attitudes towards alternative bus technologies are affected by organisational factors related to both governance frameworks (e.g., ownership type, managerial structure) and culture and values (e.g., corporate social responsibility, innovation approach).

First, transport providers' *ownership and managerial structures* may influence their ability and propensity to transition to a ZEB fleet. Various studies have found that services that are publicly owned are more likely to adopt ZEBs, due to state-wide regulations and fewer budget constraints [32,73,89]. Also the managerial structure (i.e., autocratic, bureaucratic, democratic, hierarchic) may affect the decision-making process and thus the likelihood of adopting new technologies. Mohammed et al. [28] expanded on this by showing that hierarchic organisations are more likely to be early adopters of alternatively powered buses.

*Organisation size* is another key determinant of ZEB fleet deployment. Large transit providers tend to have greater financial availability, more operational capabilities (i.e., a large fleet allows vehicle assignments to be rotated), lower energy costs, more opportunities to affect market developments, more familiarity with innovation processes, and more environmentally-friendly brand images [28,80,90,91].

Focusing on organisational culture, fleet owners' *innovation approach and risk appetite* may represent both barriers to and enablers of a ZEB transition. The literature reports an interesting trade-off related to risk appetite: on the one hand, early adopters of ZEBs risk being "guinea pigs" for technological development [80] by investing in assets that are still developing (and could be soon obsolete); however, they may benefit from a first-mover advantage [73]. Miles and Potter [92] claimed that new risk management strategies are important for a successful transition, and Aldenius et al. [32] highlighted that stakeholders face the challenge of managing unexpected outcomes related to innovation. In this regard, lock-in effects and "inertia" linked to the tendency to purchase known and mature technologies may hinder the transition to ZEBs [93,94].

*Corporate social responsibility and brand image* relate to ZEBs' positive impact on environmental stewardship, which may contribute to a good public image and increase employee morale and efficiency [54]. Indeed, corporate social responsibility is considered a primary motivator for the adoption of clean technologies [30].

Finally, the transition to new technologies may be influenced by the

*personal interests of decision-makers*, which may be linked to a wide variety of attitudes, including technophilia, scepticism/conservatism, social pressure, and lack of knowledge [18,28,33].

#### 3.3.2. Availability of reliable information and knowledge

The mass deployment of technology may be challenged by inadequate levels of information and knowledge among stakeholders.

*Performance uncertainty* is a key barrier to ZEB fleet transition. This implies a lack of reliable knowledge with respect to various service dimensions (e.g., the useful life and driving range of electric batteries, alternative charging infrastructure options and development, maintenance costs, electric system failures, R&D, environmental impacts [28, 30,80]). However, the robustness of the bus technology assessments can be improved by developing risk analysis methods [95,96].

While a growing number of studies are conducting *total cost of ownership analysis* of ZEBs, these studies are affected by uncertainty related to a lack of information about some model components, particularly with respect to the vehicle and infrastructure lifetime, future energy costs, maintenance costs, and resale value or disposal costs [29, 32,68]. In this regard, over the last decade, ZEB costs have been decreasing due to economies of scale and supply chain development, and further improvements are expected [34]. For instance, the costs of electrolysation have more than halved over the previous decade [61].

Moreover, the lack of *operational data in similar contexts* is still cited as a barrier to ZEB adoption. There is a need for real-world data in connection with challenging routes and large-scale implementation [7, 31]. Conversely, considerable data is available from demonstration projects, scientific reports, and case studies [28,63].

Finally, the diffusion of *common standards*, especially related to safe operational procedures, may affect transport operators' knowledge of and attitudes towards ZEB adoption [32].

# 3.4. Social dimension

The social dimension considers the public acceptance of ZEBs and their impact on the labour market, as well as the perceived quality of ZEBs. The latter is based on the assumption that public transport, given its relevance in terms of accessibility and social equity, is an important component of urban social sustainability [97].

# 3.4.1. Public acceptance and job development

Public acceptance and job development refer to local residents' opinions and perceptions of ZEBs and the effects of the ZEB transition on employment and job skills.

Positive *public acceptance* and media coverage are often cited as enablers of ZEB diffusion [18,30]. In this regard, environmental factors have been found to influence the choice of travel mode, with consumers that are more inclined to appreciate bus fleet electrification [11]. Conversely, public support for ZEBs is questioned due to uncertainties related to their environmental impact (especially with respect to energy production and battery disposal) and excessive dependence on a specific market (e.g., China [6,68]).

The large-scale deployment of ZEBs may *impact employment and job skills*. Specifically, it may pose new challenges, mostly related to the need for different skills, new performance capabilities and training programs for operational staff (e.g., drivers, maintenance workers) and managers [58,98,99]. Negative downstream consequences may be linked to the limited repair and maintenance regimes required for electric buses and the higher level of automation [6,75]. Conversely, there may be opportunities for new workplaces [100].

# 3.4.2. Bus service quality

Bus service quality relates to the quality evaluations made by bus drivers and passengers, with respect to the vehicle technology and the service provided.

The quality evaluation of the technology and service usually enables

ZEB adoption, as ZEBs have less vibration, run more smoothly and generate less noise, compared to internal combustion vehicles [101, 102]. ZEBs' on-board technology (e.g., WiFi, USB charging ports) and vehicle management tools are also evaluated positively [29,85], though they may give rise to cybersecurity risks [33]. Nevertheless, the frequent charging needs of electric buses can lead to service degradation [54].

Multiple ZEB characteristics positively impact *employee satisfaction* (i.e., drivers). Indeed, ZEBs' better on-board environments (i.e., reduced noise and vibration) may mitigate driver fatigue and improve operational efficiency [30]. While some studies have found that drivers perceive electric vehicles as easy to use and fun to drive [28,82], others have observed that not all drivers wish to drive electric buses, due to range anxiety – also considering that driving behaviour significantly impacts battery performance [67,103].

#### 3.5. Environmental dimension

The primary goal of public policies supporting the ZEB transition is to address climate change and urban pollution within climate action planning, which increasingly needs to be developed consistently with globally accepted criteria, standards, and benchmarks [104]. This recent study also remarks the importance of setting specific, measurable, achievable, realistic and time-bound (SMART) sectoral targets, as well as integrating mitigation and adaptation actions [104]. In this context, the proposed framework includes in the environmental dimension both the emissions of the power source and the externalities related to the entire life-cycle of vehicles and infrastructure.

# 3.5.1. Well-to-wheel emissions of the power source

Well-to-wheel emissions of the power source relate to environmental impacts due to energy/fuel production and distribution (i.e., well-totank) and tailpipe emissions (i.e., tank-to-wheel). In this respect, there are two main categories of harmful pollutants: (a) greenhouse gas emissions (GHG), such as carbon dioxide and methane, which lead to global warming and climate change, and (b) air pollutants, such as particulate matter, nitrogen oxides, and sulphur dioxide, which affect human health and surrounding habitats.

*GHG emissions* are a controversial issue in relation to ZEB diffusion. The major advantage of ZEB diffusion lies in the emissions abatement during the service provision (i.e., at the tank-to-wheel stage) [78]. However, studies have raised concerns about the production and distribution of electricity and hydrogen, which often relies on fossil sources [61,105]. A growing share of renewable energies is enabling ZEB adoption [7,59]. To this regard, there are different solutions that can help to integrate renewable energy and electric vehicles based on smart charging approaches [106].

Air pollution and marginal effects of harmful emissions are strongly in favour of ZEBs [87], as are reductions in noise pollution [30]. ZEBs effectively eliminate harmful emissions of exhaust gases and traditional engine noise. Given that commuter's personal exposure to air pollutants is higher near bus stops [107], the net benefit of the bus fleet electrification may be even greater than expected. However, some side effects should be monitored: Rodrigues and Seixas [29] shed light on the risk of toxic and flammable substance leaks; and Thorne et al. [67] noted that noise improvements may be compromised by HVAC systems and more noticeable charging operations.

## 3.5.2. Life-cycle assessments of vehicles and infrastructure

Life-cycle assessments of vehicles and infrastructure have received little attention from policy-makers, managers and scientific literature, as highlighted by recent reviews on electric buses [7,25]. Such assessments aim at measuring the cradle-to-grave life-cycle impacts of vehicle components and infrastructures, mostly linked to raw materials (i.e., production) and the end-of-life stage (i.e., waste management). In this context, some studies have raised the issue of environmental pollution generated by *vehicle and infrastructure production* and the *waste*  *management* of vehicle components. Logan et al. [59] asserted that ZEB infrastructure requires a wide range of rare raw materials, whose extraction and export may damage the environment. Equally, the effective reuse and recycling of electric batteries and their compounds remain uncertain [31]. There emerges the need for a wider adoption of circular economy principles in transport sector electrification, so far mainly focused on battery repurposing and remanufacturing [108]. However, other best practices related to vehicle ownership management can contribute to increase the level of circularity in transport industry [109].

# 3.6. Institutional dimension

Transit services rely heavily on government institutions, which, in addition to providing public subsidies, also influence market structures and technological developments through targeted policy initiatives and economic regulation.

# 3.6.1. Political will and official commitment

The degree of public intervention is closely linked to political will and the commitment of public authorities, which, in turn, depend on policy goals, national policies and other relevant markets.

Over the last decade, *policy goals and external pressure* have supported the transition to ZEBs. Multiple global and national policies currently promote the adoption of clean vehicles, thereby enabling the adoption of ZEBs [25,110]. In this regard, private fleet owners have expressed concern about political uncertainty and instability [54,111]. Moreover, some policies may hinder upstream competition and slow the market penetration of new technologies (e.g., the Buy America mandate, see Ref. [94]).

Consequently, a coherent and stable *national policy design* is crucial for large-scale ZEB deployment. Local and national policies should be coordinated through a top-down approach, as individual and local initiatives are not likely to generate sufficient impact [77,80,112]. Additionally, integrated urban plans that involve both transport and energy sectors are increasingly relevant [113,114]. Planning should not be static, but subject to systematic review [32]. Board and executive leadership should share knowledge among stakeholders and develop a network of actors to tackle industry issues [53,58,70]. Finally, long-term investment in other technologies (e.g., biogas) may affect (and be affected by) ZEB implementation; given that public transport cannot be powered by only one traction system, a clear direction.

Impacts on/from other markets linked to ZEB diffusion may also influence policy-makers and other high-level decision-makers. Specifically, the large-scale deployment of electric buses may negatively or positively affect electricity grid performance and resilience [29,58]. ZEBs may also contribute to the growth of renewable energies by implementing distributed generation hubs and reducing dependence on fossil fuel imports [54,115]. Finally, at present, the market for electric batteries is heavily dependent on Chinese industry, which is highly exposed to political interference [33].

#### 3.6.2. Policy initiatives

Public interventions generally aim at supporting ZEB diffusion through a variety of targeted policies.

*Public incentives* are the most common, aimed at reducing purchasing costs relative to conventional diesel buses, and thereby enabling ZEB adoption. Such incentives include capital subsidies (e.g., cash grants to purchase vehicles and infrastructure), tax breaks (e.g., lower registration fees) and direct investment in physical assets (e.g., charging/refuelling stations). Gallo [58] suggested adaptations to the electricity cost structure to accelerate the transition to electric buses (e.g., to remove the grid fee). However, transport operators claim that financial incentives do not make ZEB adoption economically convenient in some contexts [30]. Rodrigues and Seixas [29] asserted that new international sources of climate finance (e.g., Green Climate Fund) may play an

important role in increasing funding.

Government actions also depend on *legislative directives and requirements*, such as harmful emission quotas, low-emission zones, vehicle emission limits, and emission-base taxes [61,70,82]. A lack of uniform standards may risk deterring the large-scale deployment of ZEBs [32,99].

Finally, "*in-kind contributions*" [77] from public authorities represent an important ZEB enabler. The literature proposes various examples of these initiatives: pilot projects, information on policy initiatives, support for R&D (e.g., update studies), support for knowledge sharing (e.g., real-world operational data, streamlined guidelines, tutored vehicle trials, formalisation of tacit knowledge, lessons learned and best practices), operator assistance for defining a monitoring plan and training employees, education and awareness campaigns for users, and the introduction of comfortable ticketing systems.

However, public subsidies sometimes remain unused, mainly due to ineffective funding *arrangements*. Blynn and Attanucci [53] noted that it is difficult to comply with public funding schemes based on a fixed amount of money, since buses and infrastructures are complex investments that rarely align with the budget. Aldenius et al. [32] described that, when subsidies are provided after an operator has purchased a ZEB, the operator may not be able to anticipate the payment. Mohammed et al. [28] asserted that purchasing policies restricted to the choice of options from a predetermined list may disincentivise fleet owners who want to forge partnerships with suppliers. Du et al. [76] highlighted the risk of over-subsidising a specific technology without defining an overall strategy based on desired service features. Finally, Miles and Potter [92] concluded that subsidies, alone, are insufficient and that more is needed to support risk management approaches and the organisational structure of public transport.

#### 3.6.3. Regulation and industrial organisation

Regulation and industrial organisation focus on the connection between the economic regulatory framework and the ZEB supply chain.

Institutional innovation is needed to enable appropriate market structures and business models for the effective implementation of ZEBs [68]. In this regard, policy-makers face various challenges. First, the adoption of ZEBs in public transport requires a different division of responsibilities between operators and local authorities [32,68]. A complex mix of stakeholders (e.g., energy providers, bus manufacturers) must interact to provide the service [32,67]. Moreover, new market entrants (e.g., charging infrastructure providers) may have little understanding of public transport but hope to profit from their services [33]. There is a clear need for innovative regulation to deal with different stakeholders, ensuring an appropriate distribution of risks and promoting cooperation at all stages, from the planning phase onward [29,34,77]. Hensher [33] proposed a collaborative approach (i.e., "supply chain partnership contract") consisting of a contract or management agreement between a public authority and consortiums that account for the entire supply chain, similar to a public-private partnership. Miles and Potter [92] described a business model in which an "enabling company" purchases buses and chargers and leases them to transport operators.

In this context, the *concession terms and conditions* defined by public authorities must implement an appropriate incentive scheme for all stakeholders. Some elements of current contract rules may affect the smooth transition to ZEBs (see Refs. [67,68,87]): cost efficiency targets and acceptance of potential cost increases, contract length (i.e., the longer lifetime of ZEBs and their infrastructure relative to conventional buses), the importance of environmental criteria, arrangement flexibility (i.e., ZEBs may have start-up problems), incentives for the service provider to innovate (i.e., not only at the bidding/negotiation stage) and rights allocation at the end of the concession period.

# 4. Assessment of barriers and enablers

This section presents and discusses the results of the quantitative analysis according to the hierarchical structure of the ZEB fleet transition framework (Section 3).

# 4.1. Systematic collection of barriers and enablers and connection with the holistic framework

The second step of the content analysis was developed by systematically accounting for the barriers to and enablers of ZEB adoption identified in the literature. At this stage, 768 items (i.e., 384 barriers, 384 enablers) were manually recorded through an in-depth reading of the selected articles. These barriers and enablers were connected to one or more sub-factors of the proposed holistic framework. The result was an interaction matrix consisting of 46 columns (i.e., the sub-factors of the ZEB fleet transition framework) and 768 rows (i.e., the barriers and enablers extracted from the reviewed literature), where the cell value is 1 is the sub-factor is affected by the item, and 0 if not. Consistently with hierarchical structure of the framework, the columns of the matrix can be aggregated in the 15 decision-making factors and in turn in the 6 evaluative dimensions. This matrix constituted the analytical basis for conducting quantitative analysis on the frequency with which each decision-making factor/sub-factor was identified as a barrier or enabler in the reviewed literature. These descriptive statistics allow drawing an overview of the impact – intended as influence direction (i.e., hindering or enabling) - of the decision-making factors included in the framework on the large-scale deployment of ZEBs, as well as taking into account the effects of technology progress and improved managerial practices.

# 4.2. Impact of the decision-making factors

The first main finding referred to the contextuality of the deterring or motivating effect on ZEB adoption. The analysis collected information on the same barriers and enablers in different economic and cultural contexts, and different time periods. Some factors were considered barriers in some studies and enablers in others, depending on the geographical context, the national governance, the public transport organisation, the degree of technological development and the relevant stakeholders. This implies that there are continuous trade-offs between conflicting multi-criteria dimensions in the decision-making process, which might be strongly affected by local circumstances. For instance, Aldenius et al. [32] noted significant differences in the challenges related to electric bus adoption between four cities (in UK and Sweden), while Barfod et al. [82] found a mismatch between fleet managers and policy-makers regarding the importance of some factors. Mahmoud et al. [26] assessed techno-economic competitiveness of different electric powertrain options adopting a multi-criteria approach.

Figs. 3 and 4 show the impact of each evaluative dimension and the related factors and sub-factors, based on the reviewed literature.

ZEBs' low technological performance is mainly due to their decreased flexibility and corresponding greater operational complexity. All of the reviewed studies identified the limited driving range of battery electric buses as a critical issue, in addition to the need to recharge (both overnight and during service). This implies a greater need to define service scheduling to achieve the correct trade-off between bus frequency, recharging time and impact on the electricity grid. In this regard, the other two major barriers that were identified in the literature relate to the availability and reliability of supporting infrastructures and energy. With respect to this last point, the main challenge refers to ZEBs' high-power loads. After the grid connection is built, the electric market is less volatile than the fossil fuel market. Service features and the geographical context may hinder or facilitate the transition to ZEBs, according to local conditions.

With regard to the economic sphere, the high capital costs related to vehicle purchasing, infrastructure development and other initial costs



□ Impact as barrier □ Impact as enabler

**Fig. 3.** Impact of the evaluative dimensions and related factors<sup>\*</sup>. \* The numerical scale indicates the total number of mentions as barrier/enabler of each dimension/ factor (normalised to 100). For instance, the factor related to the Technical/operational performance was mentioned as a barrier in 128 instances (73%) and as an enabler in 47 instances (27%). The radar chart is scaled based on the most cited dimension/factor in order to provide a graphical view of the relative frequency with which the influencing factors are mentioned in the literature. Note that the total number of mentions of each dimension is the sum of the mentions of its component factors, which in turn is the sum of the references of the related sub-factors (Fig. 4).

(including indirect costs) represent another major barrier to ZEB adoption. This cost structure makes investment in ZEB fleets risky, due to the long payback period and the relative immaturity of the technology (especially with respect to fuel cell buses). The cost-effectiveness of battery electric buses (in terms of operational costs), mainly due to lower energy costs, plays an important enabling role by compensating for the higher initial expenses. Moreover, alternative debt instruments for fleet owners (e.g., green bonds, leasing schemes) may contribute to



Fig. 4. Impact of each factor and related sub-factors<sup>\*</sup>. \* The numerical scale indicates the total number of mentions as barrier/enabler of each sub-factor (normalised to 100). For instance, the sub-factor related to the Driving range was mentioned as a barrier in 51 instances (78%) and as an enabler in 14 instances (22%).

reducing the impact of the high upfront costs, even if these were not frequently cited in the literature.

The managerial dimension is influenced by two opposing factors: the lower availability of reliable information (identified as an influential deterrent to ZEB diffusion) and the organisational structure and culture of stakeholders (often cited as an enabling factor). Regarding the latter, transport operators often refer to corporate social responsibility as one of the main reasons for their switch to ZEBs. Organisational size may also play an important role by influencing financial availability, familiarity with innovation processes and attention to the company image. Indeed, many fleet owners stress the importance of being large enough to effectively manage this transition process. The fact that these aspects were highlighted more frequently as enablers than barriers suggests a likely bias in this stream of the literature, which is predominantly focused on case studies of adoption, rather than non-adoption decisions. Consequently, the participating managers reported their organisational environment, innovation approach and personal interest as facilitating factors. However, a closer examination of non-adoption decisions may find that these factors can also play the opposite role.

Conversely, knowledge about ZEB technologies was identified as an important barrier to their deployment, mostly due to uncertainties related to operational performance (e.g., battery durability and lifetime, charging procedures, maintenance costs) and thus the total cost of ownership. Also, a lack of common standards often negatively impacted adoption decisions. Practical operational data are growing, but only based on pilot projects and cases of small-scale implementation; further research on the effects of large-scale deployment is needed.

While the social dimension was the least studied in the literature related to alternative bus technologies, it is gaining importance in decision-making processes, especially from a public policy perspective. The factors related to social criteria seemed to support ZEBs. These included public acceptance (linked to ZEBs' green image), employee satisfaction (related to driver-friendliness) and service quality (e.g., onboard comfort). However, the large-scale adoption of ZEBs may have a disruptive impact on staff skills and competencies, especially with regard to maintenance and depot operations; this impact may both negatively and positively affect employment levels.

The environmental dimension represents the main motivation for switching to ZEBs (i.e., to reduce road transport pollution). The tank-towheel emissions of alternative bus powertrains (GHGs and air pollutants) have been extensively studied in both academic and grey literature. While there is no doubt that ZEBs cut harmful emissions from the tailpipe, their carbon footprint related to the well-to-tank stage depends on the source of their electricity or hydrogen energy. Nevertheless, the overall reduced environmental impact of the power technology is considered an enabler of the transition to ZEBs, also considering their reduced noise pollution. Conversely, life-cycle assessments of ZEBs show negative downstream consequences due to the wide use of raw material sources that are rare and difficult to dispose of or recycle. However, the precise impact of this latter factor is unknown, as there is little research on it.

Over and above the environmental dimension, the institutional dimension has been the most mentioned enabler for ZEB adoption. Most of the reviewed studies focused on the decision-making processes related to ZEB deployment, made by fleet owners (mostly private transport operators or transit agencies). These most often referred to policy initiatives and the need to comply with environmental regulations, rather than the perceived importance of reducing emissions (i.e., corporate social responsibility). In this framework, policy goals and external pressure, coordinated national policies, public incentives, legislative directives and in-kind contributions may play a crucial role in promoting ZEB diffusion. Of note, policies must be effectively designed and planned in order to achieve targets. However, in some cases, funding arrangement schemes were identified as barriers to ZEB adoption, making it difficult to access grants or cover expenses other than the purchase of vehicles. The enabling impact of institutional criteria was less evident with respect to economic regulation and industrial organisation. Indeed, the need for new collaborative governance frameworks, new divisions of responsibilities and risks between stakeholders, and new concession terms and conditions related to public transport services provided by ZEBs has not been consistently met by local authorities. The reviewed studies presented several examples of ZEB deployments in which policy-makers set up appropriate market structures to support the development of new business models. However, these factors were not considered in many decision-making processes, thereby limiting the adoption of ZEBs. In this respect, research on regulatory conditions that may facilitate the large-scale implementation of ZEBs is needed.

In conclusion, the transition to a ZEB fleet may be hampered by technological, economic, and managerial factors. Conversely, social, environmental, and institutional criteria are likely to support the adoption of alternative bus technologies. The hindering impact of technological, economic, and managerial dimensions explains why transport providers, who are subject to budget constraints, are often reluctant to switch to ZEB fleets. In contrast, new entrants to the industry, such as energy and infrastructure providers (other than bus manufacturers), tend to look for opportunities related to ZEB diffusion with interest. Public authorities are promoting ZEB diffusion to achieve sustainable development goals, particularly within environmental and social dimensions. However, they are challenged in their efforts to create incentive systems that facilitate the necessary market structure and business models for ZEB fleets.

# 4.3. Time trend effects on decision-making factors

Technology and practices improve over time, due to technological advancements and the typical learning curve that applies to the production and use of products. With respect to the ZEB fleet transition, this progress may involve all evaluative dimensions (e.g., increased electric battery performance, reduced costs, greater availability of information, greater knowledge about environmental protection, a growing share of renewable energies, new forms of public-private partnerships). It follows that unmet functionality, availability, and/or economic issues may be based on outdated assessments.

This review included articles published over a 25-year time span. Within this time, the relevance of many factors changed, according to technological developments. However, the sample of papers was not uniformly distributed: more than half referred to the last 3 years (52% in the period 2019-2022) and relatively few were published prior to 2008 (8%). To analyse the time trend effects on factors influencing the transition to ZEBs, the reviewed studies were split into two groups: (1) those published before and in 2018 and (2) those published after 2018. Subsequently, the potential differences in the reported impact of each factor between groups were investigated. There are two reasons behind the choice of 2018 as the dividing point: first, it could be considered the breakthrough year for ZEBs in Europe: in 2019, new electric buses registrations tripled, compared to the previous year; and the total number of ZEBs went from approximately 1650 vehicles (2018) to more than 3660 (2019) and up to 9500 (2022) in the EU [16]; second, it split the set of articles in half, illustrating how research in this area has grown in line with the market share and policy commitment to ZEBs.

The comparison between the two sets of studies was based on the frequency with which each sub-factors was mentioned as barrier and enabler in relation to the total number of mentions of each sub-factor respectively (Fig. 5).

The overall picture showed an optimistic trend, considering studies published after 2018, with barriers decreasing and enablers increasing



Fig. 5. Comparison of the percentage share of mentions as barriers and enablers for each dimension/factor up to and after 2018\*. \* The numerical scale indicates the percentage of mentions as a barrier and enabler of each dimension/factor.

in number of mentions by the literature (except for those within the environmental dimension).

Progress related to electric battery durability seemed to significantly reduce the negative impact of operational performances related to electrified bus fleets, leading to a general improvement in the technological dimension evidenced in the studies published after 2018 (even though electric battery durability still remained a main barrier to ZEBs adoption). However, focusing on specific factors, deterrents associated with infrastructure, depot operations and service scheduling seemed to increase in impact. This was expected, given the higher operational complexity of both service provision and charging management, and the lack of electricity power (i.e., critical issues linked to the switch from small-scale to large-scale ZEB deployment).

The economic dimension showed a promising trend, with the impact of CAPEX and other initial costs as barriers to ZEB adoption decreasing. This suggests that, over time, fleet owners have considered investment in these technologies less risky. In addition, the enabling impact of lower energy costs rose, perhaps due to the escalation in fossil fuel costs.

The most noticeable improvement referred to the managerial dimension, which demonstrated a reverse in the influence direction after 2018. It is likely that the growing availability of operational data and shared experiences reduced the relevance of barriers related to the reliability of information. Moreover, Corporate social responsibility became more important within organisations after 2018, thereby increasing the impact of organisational structure and the culture of transport operators as enabling factors.

The social dimension also grew in relevance, typically in support of ZEBs. Indeed, ZEBs increasingly earned the appreciation of bus passengers and drivers, who developed greater confidence in the ZEB service supply and connected ZEBs with healthy urban living.

Conversely, some downstream consequences emerged in relation to the environmental impact of ZEBs – mainly linked to the well-to-tank stage of electricity/hydrogen supply (i.e., energy production and distribution) and the manufacturing and disposal of some vehicle components (i.e., electric batteries, fuel cells). Consequently, after 2018, fewer studies highlighted only the environmental benefits of ZEBs, and the environmental dimension assumed an overall less enabling impact.

Finally, the institutional dimension remained the main motivating factor for ZEB adoption. Policy initiatives aimed at reducing harmful emissions from road transport vehicles increased in number, and studies in both groups reported the efforts of local authorities to deal with the large-scale implementation of ZEBs.

# 5. Conclusions

This research aimed at building a comprehensive framework to assess large-scale deployment of ZEBs. In line with the most common theories on the diffusion of technological and environmental innovations, the result was a hierarchical structure consisting of 6 evaluative dimensions, 15 decision-making factors, and 46 sub-factors. It emerged that the decision-making process involves continuous tradeoffs between the multi-criteria dimensions, which are also affected by local contexts and stakeholder perspectives. In a second step, the barriers and enablers identified in the reviewed literature were quantitatively analysed to assess the influence direction of each factor on ZEB diffusion. The results showed that technological, economic, and managerial dimensions have hindered the large-scale deployment of ZEBs, while social, environmental, and institutional factors have stimulated ZEB adoption. The time trend analysis showed an optimistic trend due to technological progress and scale economies related to ZEBs' operational and economic performance (with respect to, e.g., batteries and related infrastructure).

Based on the proposed framework, the following topics are of increasing importance for the transition to ZEBs and not fully addressed in the literature. Therefore, future research should aim at exploring these areas, in depth.

- Public transport revenues: investigating whether and how the adoption of ZEBs might affect public transport demand, which is often the first source of funding for bus operators and a key enabler of investment in ZEB technologies.
- Life-cycle assessment of buses, components, and related infrastructure to enable circular business models in transport sector: evaluating the environmental impact of ZEB technologies, also considering externalities related to the production and disposal of vehicles, their components (e.g., electric batteries, fuel cells), and supporting infrastructures.
- Mixed fleet management: exploring the transition to ZEBs is a gradual process, during which transport operators may face the challenge of simultaneously managing different power technologies, often at the same depot. This may introduce greater complexity, also with respect to staff skills and competencies.
- Regulatory framework: assessing how different funding policies and regulation may impact the development of appropriate market structures and business models for large-scale ZEB deployment. In this context, collaboration among stakeholders and flexibility in transport planning are crucial to manage uncertainties concerning the transition.

The proposed framework may support both policy-makers and private operators in their efforts to holistically plan the transition to ZEBs, providing an overview of the main conflicting criteria that should be addressed and managed. Moreover, it represents a starting point for conducting multi-criteria assessments of alternative bus technologies or comparing multiple case studies of fleet renewal projects. Finally, extending the analysis to encompass both private vehicles and freight transport opens new possibilities for comprehensive assessments of transport decarbonisation. The integrated study of electrification across private vehicles, freight transport, and public transport emerges as a promising avenue of research.

# Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

# Data availability

Data will be made available on request.

# References

- IEA International Energy Agency. Global EV outlook 2022. https://W wwIeaOrg/Reports/Global-Ev-Outlook-2022; 2022.
- [2] Muratori M, Jadun P, Bush B, Bielen D, Vimmerstedt L, Gonder J, et al. Future integrated mobility-energy systems: a modeling perspective. Renew Sustain Energy Rev 2020;119. https://doi.org/10.1016/j.rser.2019.109541.
- [3] Hawkins TR, Singh B, Majeau-Bettez G, Strømman AH. Comparative environmental life cycle assessment of conventional and electric vehicles. J Ind Ecol 2013;17:53–64. https://doi.org/10.1111/j.1530-9290.2012.00532.x.
- [4] Bartolozzi I, Rizzi F, Frey M. Comparison between hydrogen and electric vehicles by life cycle assessment: a case study in Tuscany, Italy. Appl Energy 2013;101: 103–11. https://doi.org/10.1016/j.apenergy.2012.03.021.
- [5] Alali L, Niesten E, Gagliardi D. The impact of UK financial incentives on the adoption of electric fleets: the moderation effect of GDP change. Transport Res Part A Policy Pract 2022;161:200–20. https://doi.org/10.1016/j. tra.2022.04,011.
- [6] Berkeley N, Bailey D, Jones A, Jarvis D. Assessing the transition towards battery electric vehicles: a multi-level perspective on drivers of, and barriers to, take up. Transport Res Part A Policy Pract 2017;106:320–32. https://doi.org/10.1016/j. tra.2017.10.004.
- [7] Eldeeb G, Mohamed M. Transit electrification state of the art: a machine-learning based text mining approach. Transp Res D Transp Environ 2022;111. https://doi. org/10.1016/j.trd.2022.103446.
- [8] Tong F, Hendrickson C, Biehler A, Jaramillo P, Seki S. Life cycle ownership cost and environmental externality of alternative fuel options for transit buses. Transp Res D Transp Environ 2017;57:287–302. https://doi.org/10.1016/j. trd.2017.09.023.

#### A. Avenali et al.

- [9] Nesbitt K, Sperling D. Myths regarding alternative fuel vehicle demand by lightduty vehicle fleets. Transp Res D Transp Environ 1998;3:259–69.
- [10] van der Straten P, Wiegmans BW, Schelling AB. Enablers and barriers to the adoption of alternatively powered buses. Transport Rev 2007;27:679–98. https:// doi.org/10.1080/01441640701248518.
- [11] Sunitiyoso Y, Belgiawan PF, Rizki M, Hasyimi V. Public acceptance and the environmental impact of electric bus services. Transp Res D Transp Environ 2022; 109. https://doi.org/10.1016/j.trd.2022.103358.
- [12] Tan R, Lin B. Public perception of new energy vehicles: evidence from willingness to pay for new energy bus fares in China. Energy Pol 2019;130:347–54. https:// doi.org/10.1016/j.enpol.2019.04.010.
- [13] Lin B, Tan R. Are people willing to pay more for new energy bus fares? Energy 2017;130:365–72. https://doi.org/10.1016/j.energy.2017.04.153.
- [14] Pagliaro M, Meneguzzo F. Electric bus: a critical overview on the dawn of its widespread uptake. Adv Sustain Syst 2019;3. https://doi.org/10.1002/ adsu.201800151.
- [15] ACEA European Automobile Manufacturers Association. The automobile industry pocket guide 2022–2023. https://WwwAceaAuto/Publication/Automo bile-Industry-Pocket-Guide-2022-2023/2022.
- [16] EAFO European alternative fuels observatory. Alternative fuels in Europe. htt ps://Alternative-Fuels-ObservatoryEcEuropaEu/2022.
- [17] Singh V, Singh V, Vaibhav S. A review and simple meta-analysis of factors influencing adoption of electric vehicles. Transp Res D Transp Environ 2020;86. https://doi.org/10.1016/j.trd.2020.102436.
- [18] Biresselioglu ME, Demirbag Kaplan M, Yilmaz BK. Electric mobility in Europe: a comprehensive review of motivators and barriers in decision making processes. Transport Res Part A Policy Pract 2018;109:1–13. https://doi.org/10.1016/j. tra.2018.01.017.
- [19] Coffman M, Bernstein P, Wee S. Electric vehicles revisited: a review of factors that affect adoption. Transport Rev 2017;37:79–93. https://doi.org/10.1080/ 01441647.2016.1217282.
- [20] Li W, Long R, Chen H, Geng J. A review of factors influencing consumer intentions to adopt battery electric vehicles. Renew Sustain Energy Rev 2017;78: 318–28. https://doi.org/10.1016/j.rser.2017.04.076.
- [21] Adnan N, Nordin SM, Rahman I, Vasant PM, Noor A. A comprehensive review on theoretical framework-based electric vehicle consumer adoption research. Int J Energy Res 2017;41:317–35. https://doi.org/10.1002/er.3640.
- [22] Rezvani Z, Jansson J, Bodin J. Advances in consumer electric vehicle adoption research: a review and research agenda. Transp Res D Transp Environ 2015;34: 122–36. https://doi.org/10.1016/j.trd.2014.10.010.
- [23] Mohamed M, Higgins C, Ferguson M, Kanaroglou P. Identifying and characterizing potential electric vehicle adopters in Canada: a two-stage modelling approach. Transport Pol 2016;52:100–12. https://doi.org/10.1016/j. tranpol.2016.07.006.
- [24] Mahmoud M, Garnett R, Ferguson M, Kanaroglou P. Electric buses: a review of alternative powertrains. Renew Sustain Energy Rev 2016;62:673–84. https://doi. org/10.1016/j.rser.2016.05.019.
- [25] Manzolli JA, Trovão JP, Antunes CH. A review of electric bus vehicles research topics – methods and trends. Renew Sustain Energy Rev 2022;159. https://doi. org/10.1016/j.rser.2022.112211.
- [26] Mahmoud M, Garnett R, Ferguson M, Kanaroglou P. Electric buses: a review of alternative powertrains. Renew Sustain Energy Rev 2016;62:673–84. https://doi. org/10.1016/j.rser.2016.05.019.
- [27] Barman P, Dutta L. Charging infrastructure planning for transportation electrification in India: a review. Renew Sustain Energy Rev 2024;192:114265. https://doi.org/10.1016/j.rser.2023.114265.
- [28] Mohammed L, Niesten E, Gagliardi D. Adoption of alternative fuel vehicle fleets a theoretical framework of barriers and enablers. Transp Res D Transp Environ 2020;88. https://doi.org/10.1016/j.trd.2020.102558.
- [29] Rodrigues ALP, Seixas SRC. Battery-electric buses and their implementation barriers: analysis and prospects for sustainability. Sustain Energy Technol Assessments 2022:51. https://doi.org/10.1016/j.seta.2021.101896.
- [30] Bae Y, Mira SK, Rindt CR, Ritchie SG. Factors influencing alternative fuel adoption decisions in heavy-duty vehicle fleets. Transp Res D Transp Environ 2022;102. https://doi.org/10.1016/j.trd.2021.103150.
- [31] Papa G, Santo Zarnik M, Vukašinović V. Electric-bus routes in hilly urban areas: overview and challenges. Renew Sustain Energy Rev 2022;165. https://doi.org/ 10.1016/j.rser.2022.112555.
- [32] Aldenius M, Mullen C, Pettersson-Löfstedt F. Electric buses in England and Sweden – overcoming barriers to introduction. Transp Res D Transp Environ 2022;104. https://doi.org/10.1016/j.trd.2022.103204.
- [33] Hensher DA, Wei E, Balbontin C. Comparative assessment of zero emission electric and hydrogen buses in Australia. Transp Res D Transp Environ 2022;102. https://doi.org/10.1016/j.trd.2021.103130.
- [34] Borén S, Grauers A. Stakeholder collaboration models for public transport procurement of electric bus systems. Int J Sustain Pol Pract 2019;15. https://doi. org/10.18848/2325-1166/CGP.
- [35] Bakker S, Maat K, van Wee B. Stakeholders interests, expectations, and strategies regarding the development and implementation of electric vehicles: the case of The Netherlands. Transport Res Part A Policy Pract 2014;66:52–64. https://doi. org/10.1016/j.tra.2014.04.018.
- [36] Brito TLF, Moutinho dos Santos E, Galbieri R, Costa HK de M. Qualitative Comparative Analysis of cities that introduced compressed natural gas to their urban bus fleet. Renew Sustain Energy Rev 2017;71:502–8. https://doi.org/ 10.1016/j.rser.2016.12.077.

- Renewable and Sustainable Energy Reviews 197 (2024) 114388
- [37] Wohlin C. Guidelines for snowballing in systematic literature studies and a replication in software engineering. In: ACM international conference proceeding series. Association for Computing Machinery; 2014. https://doi.org/10.1145/ 2601248.2601268.
- [38] Eisenhardt KM, Graebner ME, Sonenshein S. Grand challenges and inductive methods: rigor without rigor mortis. Acad Manag J 2016;59:1113–23. https:// doi.org/10.5465/amj.2016.4004.
- [39] Holland SP, Mansur ET, Muller NZ, Yates AJ. The environmental benefits of transportation electrification: urban buses. Energy Pol 2021;148. https://doi.org/ 10.1016/j.enpol.2020.111921.
- [40] Gustafsson M, Svensson N, Anderberg S. Energy performance indicators as policy support for public bus transport – the case of Sweden. Transp Res D Transp Environ 2018;65:697–709. https://doi.org/10.1016/j.trd.2018.10.008.
- [41] Kumar A, Sah B, Singh AR, Deng Y, He X, Kumar P, et al. A review of multi criteria decision making (MCDM) towards sustainable renewable energy development. Renew Sustain Energy Rev 2017;69:596–609. https://doi.org/10.1016/j. rser.2016.11.191.
- [42] Macharis C, de Witte A, Ampe J. The multi-actor, multi-criteria analysis methodology (MAMCA) for the evaluation of transport projects: theory and practice. J Adv Transport 2009;43:183–202. https://doi.org/10.1002/ atr.5670430206.
- [43] Tzeng GH, Lin CW, Opricovic S. Multi-criteria analysis of alternative-fuel buses for public transportation. Energy Pol 2005;33:1373–83. https://doi.org/ 10.1016/j.enpol.2003.12.014.
- [44] Spangenberg JH. Institutional sustainability indicators: an analysis of the institutions in Agenda 21 and a draft set of indicators for monitoring their effectivity. Sustain Dev 2002;10:103–15. https://doi.org/10.1002/sd.184.
- [45] Purvis B, Mao Y, Robinson D. Three pillars of sustainability: in search of conceptual origins. Sustain Sci 2019;14:681–95. https://doi.org/10.1017/ S0376892900011449.
- [46] Tornatzky LG, Fleischer M. Processes of technological innovation. Lexington books; 1990.
- [47] Venkatesh V, Morris MG, Davis GB, Davis FD. User acceptance of information technology: toward a unified view. MIS Q 2003;27:425–78.
- [48] Rogers EM. Diffusion of innovations. Free Press; 1983.
- [49] Sovacool BK. Experts, theories, and electric mobility transitions: toward an integrated conceptual framework for the adoption of electric vehicles. Energy Res Social Sci 2017;27:78–95. https://doi.org/10.1016/j.erss.2017.02.014.
- [50] Egbue O, Long S. Barriers to widespread adoption of electric vehicles: an analysis of consumer attitudes and perceptions. Energy Pol 2012;48:717–29. https://doi. org/10.1016/j.enpol.2012.06.009.
- [51] Sharma R, Malaviya P. Ecosystem services and climate action from a circular bioeconomy perspective. Renew Sustain Energy Rev 2023;175. https://doi.org/ 10.1016/j.rser.2023.113164.
- [52] Fiack D, Cumberbatch J, Sutherland M, Zerphey N. Sustainable adaptation: social equity and local climate adaptation planning in U.S. cities. Cities 2021;115. https://doi.org/10.1016/j.cities.2021.103235.
- [53] Blynn K, Attanucci J. Accelerating bus electrification: a mixed methods analysis of barriers and drivers to scaling transit fleet electrification. Transport Res Rec 2019;2673:577–87. https://doi.org/10.1177/0361198119842117.
- [54] Demeulenaere X. The use of automotive fleets to support the diffusion of Alternative Fuel Vehicles: a Rapid Evidence Assessment of barriers and decision mechanisms. Res Transport Econ 2019;76. https://doi.org/10.1016/j. retrec.2019.100738.
- [55] Dai H, Jiang B, Hu X, Lin X, Wei X, Pecht M. Advanced battery management strategies for a sustainable energy future: multilayer design concepts and research trends. Renew Sustain Energy Rev 2021;138. https://doi.org/10.1016/j. rser.2020.110480.
- [56] Dai H, Yu C, Wei X, Sun Z. State of charge estimation for lithium-ion pouch batteries based on stress measurement. Energy 2017;129:16–27. https://doi.org/ 10.1016/j.energy.2017.04.099.
- [57] Majumder S, De K, Kumar P, Rayudu R. A green public transportation system using E-buses: a technical and commercial feasibility study. Sustain Cities Soc 2019;51. https://doi.org/10.1016/j.scs.2019.101789.
- [58] Gallo J-B. Electric truck & bus grid integration, opportunities, challenges & recommendations. World Electric Vehicle Journa 2016;8.
- [59] Logan KG, Hastings A, Nelson JD. Challenges of implementing electric and hydrogen public transport. Green Energy and Technology, Springer Science and Business Media Deutschland GmbH; 2022. p. 59–80. https://doi.org/10.1007/ 978-3-030-96674-4\_5.
- [60] Wellik TK, Griffin JR, Kockelman KM, Mohamed M. Utility-transit nexus: leveraging intelligently charged electrified transit to support a renewable energy grid. Renew Sustain Energy Rev 2021;139. https://doi.org/10.1016/j. rser.2020.110657.
- [61] Vodovozov V, Raud Z, Petlenkov E. Review of energy challenges and horizons of hydrogen city buses. Energies 2022;15. https://doi.org/10.3390/en15196945.
- [62] Qiao D, Wei X, Fan W, Jiang B, Lai X, Zheng Y, et al. Toward safe carbon-neutral transportation: battery internal short circuit diagnosis based on cloud data for electric vehicles. Appl Energy 2022;317. https://doi.org/10.1016/j. apenergy.2022.119168.
- [63] Deliali A, Chhan D, Oliver J, Sayess R, Godri Pollitt KJ, Christofa E. Transitioning to zero-emission bus fleets: state of practice of implementations in the United States. Transport Rev 2021;41:164–91. https://doi.org/10.1080/ 01441647.2020.1800132.
- [64] Ma X, Miao R, Wu X, Liu X. Examining influential factors on the energy consumption of electric and diesel buses: a data-driven analysis of large-scale

public transit network in Beijing. Energy 2021;216. https://doi.org/10.1016/j. energy.2020.119196.

- [65] Chen Z, Yin Y, Song Z. A cost-competitiveness analysis of charging infrastructure for electric bus operations. Transport Res C Emerg Technol 2018;93:351–66. https://doi.org/10.1016/j.trc.2018.06.006.
- [66] He X, Zhang S, Ke W, Zheng Y, Zhou B, Liang X, et al. Energy consumption and well-to-wheels air pollutant emissions of battery electric buses under complex operating conditions and implications on fleet electrification. J Clean Prod 2018; 171:714–22. https://doi.org/10.1016/j.jclepro.2017.10.017.
- [67] Thorne RJ, Hovi IB, Figenbaum E, Pinchasik DR, Amundsen AH, Hagman R. Facilitating adoption of electric buses through policy: learnings from a trial in Norway. Energy Pol 2021;155. https://doi.org/10.1016/j.enpol.2021.112310.
- [68] Bakker S, Konings R. The transition to zero-emission buses in public transport the need for institutional innovation. Transp Res D Transp Environ 2018;64: 204–15. https://doi.org/10.1016/j.trd.2017.08.023.
- [69] De K, Majumder S, Kumar P. Analysis of the resilience of an electric-based transportation system. IET Electr Syst Transp 2020;10:23–34. https://doi.org/ 10.1049/iet-est.2018.5087.
- [70] Trencher G, Edianto A. Drivers and barriers to the adoption of fuel cell passenger vehicles and buses in Germany. Energies 2021;14. https://doi.org/10.3390/ en1404.
- [71] González LG, Cordero-Moreno D, Espinoza JL. Public transportation with electric traction: experiences and challenges in an Andean city. Renew Sustain Energy Rev 2021;141. https://doi.org/10.1016/j.rser.2021.110768.
- [72] Hensher DA. Is it time for a new bus contract procurement model under a zero emissions bus setting? Transport Res Part A Policy Pract 2022;163:80–7. https:// doi.org/10.1016/j.tra.2022.06.015.
- [73] Sierzchula W. Factors influencing fleet manager adoption of electric vehicles. Transp Res D Transp Environ 2014;31:126–34. https://doi.org/10.1016/j. trd.2014.05.022.
- [74] Xylia M, Leduc S, Patrizio P, Kraxner F, Silveira S. Locating charging infrastructure for electric buses in Stockholm. Transport Res C Emerg Technol 2017;78:183–200. https://doi.org/10.1016/j.trc.2017.03.005.
- [75] Sun Y, Wang J, Li C, Liu K. Competitiveness evaluation of electric bus charging services based on analytic hierarchy process. World Electric Vehicle Journal 2022;13. https://doi.org/10.3390/wevj13050081.
- [76] Du J, Li F, Li J, Wu X, Song Z, Zou Y, et al. Evaluating the technological evolution of battery electric buses: China as a case. Energy 2019;176:309–19. https://doi. org/10.1016/j.energy.2019.03.084.
- [77] Li X, Castellanos S, Maassen A. Emerging trends and innovations for electric bus adoption—a comparative case study of contracting and financing of 22 cities in the Americas, Asia-Pacific, and Europe. Res Transport Econ 2018;69:470–81. https://doi.org/10.1016/j.retrec.2018.06.016.
- [78] Li JQ. Battery-electric transit bus developments and operations: a review. Int J Sustain Transp 2016;10:157–69. https://doi.org/10.1080/ 15568318.2013.872737.
- [79] Wołek M, Wolański M, Bartłomiejczyk M, Wyszomirski O, Grzelec K, Hebel K. Ensuring sustainable development of urban public transport: a case study of the trolleybus system in Gdynia and Sopot (Poland). J Clean Prod 2021;279. https:// doi.org/10.1016/j.jclepro.2020.123807.
- [80] Mohamed M, Ferguson M, Kanaroglou P. What hinders adoption of the electric bus in Canadian transit? Perspectives of transit providers. Transp Res D Transp Environ 2018;64:134–49. https://doi.org/10.1016/j.trd.2017.09.019.
- [81] Wenz KP, Serrano-Guerrero X, Barragán-Escandón A, González LG, Clairand JM. Route prioritization of urban public transportation from conventional to electric buses: a new methodology and a study of case in an intermediate city of Ecuador. Renew Sustain Energy Rev 2021;148. https://doi.org/10.1016/j. rser 2021 111215
- [82] Barfod MB, Kaplan S, Frenzel I, Klauenberg J. COPE-SMARTER a decision support system for analysing the challenges, opportunities and policy initiatives: a case study of electric commercial vehicles market diffusion in Denmark. Res Transport Econ 2016;55:3–11. https://doi.org/10.1016/j.retrec.2016.04.005.
- [83] Anderhofstadt B, Spinler S. Factors affecting the purchasing decision and operation of alternative fuel-powered heavy-duty trucks in Germany – a Delphi study. Transp Res D Transp Environ 2019;73:87–107. https://doi.org/10.1016/j. trd.2019.06.003.
- [84] Avenali A, Catalano G, Giagnorio M, Matteucci G. Assessing cost-effectiveness of alternative bus technologies: evidence from US transit agencies. Transp Res D Transp Environ 2023;117. https://doi.org/10.1016/j.trd.2023.103648.
- [85] Bezruchonak A. Geographic features of zero-emissions urban mobility: the case of electric buses in Europe and Belarus. Eur Spatial Res Pol 2019;26:81–99. https:// doi.org/10.18778/1231-1952.26.1.05.
- [86] Lajunen A. Lifecycle costs and charging requirements of electric buses with different charging methods. J Clean Prod 2018;172:56–67. https://doi.org/ 10.1016/j.jclepro.2017.10.066.
- [87] Xylia M, Silveira S. On the road to fossil-free public transport: the case of Swedish bus fleets. Energy Pol 2017;100:397–412. https://doi.org/10.1016/j. enpol.2016.02.024.
- [88] Eldeeb G, Sears S, Mohamed M. What do users want from transit? Qualitative analysis of current and potential users' perceptions. Transport Res Part A Policy Pract 2023;171:103649. https://doi.org/10.1016/j.tra.2023.103649.
- [89] Golob TF, Torous J, Bradley M, Brownstone D, Soltani Crane S, Bunch DS. Commercial fleet demand for alternative-fuel vehicles in California\*, vol. 31; 1997.

- [90] Aldenius M, Khan J. Strategic use of green public procurement in the bus sector: challenges and opportunities. J Clean Prod 2017;164:250–7. https://doi.org/ 10.1016/j.jclepro.2017.06.196.
- [91] Mattson J. Use of alternative fuels and hybrids by small urban and rural transit. J Public Trans 2012;15.
- [92] Miles J, Potter S. Developing a viable electric bus service: the Milton Keynes demonstration project. Res Transport Econ 2014;48:357–63. https://doi.org/ 10.1016/j.retrec.2014.09.063.
- [93] Galván J, Cantillo V, Arellana J. Factors influencing demand for buses powered by alternative energy sources, vol. 19; 2016.
- [94] Li S, Kahn ME, Nickelsburg J. Public transit bus procurement: the role of energy prices, regulation and federal subsidies. J Urban Econ 2015;87:57–71. https:// doi.org/10.1016/j.jue.2015.01.004.
- [95] Harris A, Soban D, Smyth BM, Best R. Assessing life cycle impacts and the risk and uncertainty of alternative bus technologies. Renew Sustain Energy Rev 2018;97: 569–79. https://doi.org/10.1016/j.rser.2018.08.045.
- [96] Zhou Y, Ong GP, Meng Q, Cui H. Electric bus charging facility planning with uncertainties: model formulation and algorithm design. Transport Res C Emerg Technol 2023;150. https://doi.org/10.1016/j.trc.2023.104108.
- [97] Dempsey N, Bramley G, Power S, Brown C. The social dimension of sustainable development: defining urban social sustainability. Sustain Dev 2011;19:289–300. https://doi.org/10.1002/sd.417.
- [98] Simao JV, Cellina F, Rudel R. Critical barriers precluding the electrification of road public transport in Southern Switzerland. In: 2020 15th international conference on ecological vehicles and renewable energies, EVER 2020. Institute of Electrical and Electronics Engineers Inc.; 2020. https://doi.org/10.1109/ EVER48776.2020.9242949.
- [99] Chen F, Fernandes TRC, Yetano Roche M, da Graça Carvalho M. Investigation of challenges to the utilization of fuel cell buses in the EU vs transition economies. Renew Sustain Energy Rev 2007;11:357–64. https://doi.org/10.1016/j. rser.2005.01.007.
- [100] Tsita KG, Pilavachi PA. Evaluation of alternative fuels for the Greek road transport sector using the analytic hierarchy process. Energy Pol 2012;48:677–86. https://doi.org/10.1016/j.enpol.2012.05.079.
- [101] di Foggia G. Drivers and challenges of electric vehicles integration in corporate fleet: an empirical survey. Research in Transportation Business and Management 2021;41. https://doi.org/10.1016/j.rtbm.2021.100627.
- [102] Petrov A, Petrova D. Electric bus in Tyumen: the chase for trends and problems of exploitation; Electric bus in Tyumen: the chase for trends and problems of exploitation. E3SWebofConferences 2019. https://doi.org/10.1051/e3sconf/ 2019.
- [103] Nan S, Tu R, Li T, Sun J, Chen H. From driving behavior to energy consumption: a novel method to predict the energy consumption of electric bus. Energy 2022; 261. https://doi.org/10.1016/j.energy.2022.125188.
- [104] Aboagye PD, Sharifi A. Urban climate adaptation and mitigation action plans: a critical review. Renew Sustain Energy Rev 2024;189. https://doi.org/10.1016/j. rser.2023.113886.
- [105] Xylia M, Leduc S, Laurent AB, Patrizio P, van der Meer Y, Kraxner F, et al. Impact of bus electrification on carbon emissions: the case of Stockholm. J Clean Prod 2019;209:74–87. https://doi.org/10.1016/j.jclepro.2018.10.085.
- [106] Barman P, Dutta L, Bordoloi S, Kalita A, Buragohain P, Bharali S, et al. Renewable energy integration with electric vehicle technology: a review of the existing smart charging approaches. Renew Sustain Energy Rev 2023;183. https://doi.org/ 10.1016/j.rser.2023.113518.
- [107] Morales-Betancourt R, Wilches-Mogollon MA, Sarmiento OL, Mendez Molano D, Angulo D, Filigrana P, et al. Commuter's personal exposure to air pollutants after the implementation of a cable car for public transport: results of the natural experiment TrUST. Sci Total Environ 2023;865. https://doi.org/10.1016/j. scitotenv.2022.160880.
- [108] Schulz-Mönninghoff M, Neidhardt M, Niero M. What is the contribution of different business processes to material circularity at company-level? A case study for electric vehicle batteries. J Clean Prod 2023;382. https://doi.org/10.1016/j. jclepro.2022.135232.
- [109] Saidani M, Yannou B, Leroy Y, Cluzel F. Heavy vehicles on the road towards the circular economy: analysis and comparison with the automotive industry. Resour Conserv Recycl 2018;135:108–22. https://doi.org/10.1016/j. resconrec.2017.06.017.
- [110] Hasan MA, Chapman R, Frame DJ. Acceptability of transport emissions reduction policies: a multi-criteria analysis. Renew Sustain Energy Rev 2020;133. https:// doi.org/10.1016/j.rser.2020.110298.
- [111] Fenton P, Kanda W. Barriers to the diffusion of renewable energy: studies of biogas for transport in two European cities. J Environ Plann Manag 2017;60: 725–42. https://doi.org/10.1080/09640568.2016.1176557.
- [112] Barisa A, Rosa M, Kisele A. Introducing electric mobility in Latvian municipalities: results of a survey. Energy Proc 2016;95:50–7. https://doi.org/10.1016/j. egypro.2016.09.015. Elsevier Ltd.
- [113] Olsson L, Hjalmarsson L, Wikström M, Larsson M. Bridging the implementation gap: combining backcasting and policy analysis to study renewable energy in urban road transport. Transport Pol 2015;37:72–82. https://doi.org/10.1016/j. tranpol.2014.10.014.
- [114] Say K, Csereklyei Z, Brown FG, Wang C. The economics of public transport electrification: a case study from Victoria, Australia. Energy Econ 2023;120: 106599. https://doi.org/10.1016/j.eneco.2023.106599.
- [115] Majumder S, De K, Kumar P, Sengupta B, Biswas PK. Techno-commercial analysis of sustainable E-bus-based public transit systems: an Indian case study. Renew Sustain Energy Rev 2021;144. https://doi.org/10.1016/j.rser.2021.111033.